Check for updates

Citation: Ahmed H, Farewell D, Francis NA, Paranjothy S, Butler CC (2018) Risk of adverse outcomes following urinary tract infection in older people with renal impairment: Retrospective cohort study using linked health record data. PLoS Med 15(9): e1002652. https://doi.org/10.1371/journal. pmed.1002652

Academic Editor: Maarten W. Taal, Royal Derby Hospital, UNITED KINGDOM

Received: May 16, 2018

Accepted: August 13, 2018

Published: September 10, 2018

Copyright: © 2018 Ahmed et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data analysed for this study were obtained under institutional license from the Clinical Practice Research Datalink, https://www.cprd.com/intro.asp. Data are not available for sharing but can be applied for through the CPRD. Relevant information to allow acquisition of a replicable data set is available in the paper and its Supporting Information files.

Funding: This report is independent research arising from a National Institute of Health Research

RESEARCH ARTICLE

Risk of adverse outcomes following urinary tract infection in older people with renal impairment: Retrospective cohort study using linked health record data

Haroon Ahmed¹*, Daniel Farewell¹, Nick A. Francis¹, Shantini Paranjothy¹, Christopher C. Butler²

1 Division of Population Medicine, Cardiff University School of Medicine, Cardiff, United Kingdom, 2 Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, United Kingdom

* Ahmedh2@cardiff.ac.uk

Abstract

Background

Few studies have investigated the risk of adverse outcomes in older people with renal impairment presenting to primary care with a urinary tract infection (UTI). The aim of this study was to determine the risk of adverse outcomes in patients aged \geq 65 years presenting to primary care with a UTI, by estimated glomerular filtration rate (eGFR) and empirical prescription of nitrofurantoin versus trimethoprim.

Methods and findings

This was a retrospective cohort study using linked health record data from 795,484 patients from 393 general practices in England, who were aged \geq 65 years between 2010 and 2016. Patients were entered into the cohort if they presented with a UTI and had a creatinine measurement in the 24 months prior to presentation. We calculated an eGFR to estimate risk of adverse outcomes by renal function, and propensity-score matched patients with eGFRs <60 mL/minute/1.73 m² to estimate risk of adverse outcomes between those prescribed trimethoprim and nitrofurantoin. Outcomes were 14-day risk of reconsultation for urinary symptoms and same-day antibiotic prescription (proxy for treatment nonresponse), hospitalisation for UTI, sepsis, or acute kidney injury (AKI), and 28-day risk of death. Of 123,607 eligible patients with a UTI, we calculated an eGFR for 116,945 (95%). Median age was 76 (IQR, 70-83) years and 32,428 (28%) were male. Compared to an eGFR of >60 mL/minute/ 1.73 m², patients with an eGFR of <60 mL/minute/1.73 m² had greater odds of hospitalisation for UTI (adjusted odds ratios [ORs] ranged from 1.14 [95% confidence interval (CI) 1.01–1.28, p = 0.028], for eGFRs of 45–59, to 1.68 [95% CI 1.01–2.82, p < 0.001] for eGFRs <15) and AKI (adjusted ORs ranged from 1.57 [95% CI 1.29–1.91, p < 0.001], for eGFRs of 45–59, to 4.53 [95% CI 2.52–8.17, p < 0.001] for eGFRs <15). Compared to an eGFR of >60 mL/minute/1.73 m², patients with an eGFR <45 had significantly greater odds of hospitalisation for sepsis, and those with an eGFR <30 had significantly greater odds of

(NIHR) Doctoral Research Fellowship awarded to HA, and supported by Health and Care Research Wales (HCRW) (grant number DRF-2014-07-010). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AKI, acute kidney injury; CI, confidence interval; CPRD, Clinical Practice Research Datalink; eGFR, estimated glomerular filtration rate; GFR, glomerular filtration rate; MDRD, Modification of Diet in Renal Disease; OR, odds ratio; RECORD, Reporting of Studies Conducted using Observational Routinely-collected Health Data; UTI, urinary tract infection. death. Compared to trimethoprim, nitrofurantoin prescribing was associated with lower odds of hospitalisation for AKI (ORs ranged from 0.62 [95% CI 0.40–0.94, p = 0.025], for eGFRs of 45–59, to 0.45 [95% CI 0.25–0.81, p = 0.008] for eGFRs <30). Nitrofurantoin was not associated with greater odds of any adverse outcome. Our study lacked data on urine microbiology and antibiotic-related adverse events. Despite our design, residual confounding may still have affected some of our findings.

Conclusions

Older patients with renal impairment presenting to primary care with a UTI had an increased risk of UTI-related hospitalisation and death, suggesting a need for interventions that reduce the risk of these adverse outcomes. Nitrofurantoin prescribing was not associated with an increased risk of adverse outcomes in patients with an eGFR <60 mL/minute/1.73 m² and could be used more widely in this population.

Author summary

Why was this study done?

- It is not known if older adults with impaired kidney function are at increased risk of hospitalisation or death following a urinary tract infection (UTI).
- Nitrofurantoin is an antibiotic used to treat UTI but is not recommended in people with impaired kidney function. However, the evidence supporting this recommendation is limited.

What did the researchers do and find?

- This study used linked health record data from general practices and hospitals in England and estimated risk of hospitalisation and death for older adults with impaired kidney function presenting to primary care with a suspected UTI.
- Older adults with impaired kidney function had greater risk of a UTI-related hospitalisation and death in the 14–28 days following a UTI.
- Older adults with impaired kidney function who were treated with nitrofurantoin were not at greater risk of an adverse outcome and were less likely to experience a hospital admission for worsening kidney function.

What do these findings mean?

- There is a need for strategies that prevent UTIs and reduce the risk of UTI-related hospitalisations and death in older adults with impaired kidney function.
- Nitrofurantoin was not associated with worse outcomes and could be used more widely in this population.

Introduction

The Kidney Disease Improving Global Outcomes working group defines degrees of renal impairment using the glomerular filtration rate (GFR) [1]. GFRs <60 mL/minute/1.73 m² are split into four groups and reflect worsening renal function, from mild impairment to renal failure. Around 6% of adults in the United Kingdom have an estimated glomerular filtration rate (eGFR) of <60 mL/minute/1.73 m² [2]. This increases with age to around 20% of those aged \geq 65. There is increasing evidence of an association between renal impairment and infection [3–6]. Oxidative stress in renal impairment disrupts the function of inflammatory cytokines and may impair immune response during an infection [7]. In more severe renal impairment, uremic toxins impair the function of T-lymphocyte and antigen-presenting cells, which play important roles in cellular and humoral immunity [8]. Despite the high prevalence in older adults, and the association with infection, few studies have investigated outcomes following an infectious illness in older people with renal impairment.

A cohort study in UK primary care showed that around 20% of adults aged 65 and over had at least one urinary tract infection (UTI) over a median follow-up of 5 years [9]. Most adults presenting to primary care with symptoms and signs of a UTI receive empirical antibiotics at the same consultation, without knowledge of microbiological findings or antibiotic susceptibilities [10]. Nitrofurantoin and trimethoprim (alone or with sulfamethoxazole) are the two most commonly prescribed antibiotics for empirical treatment of UTIs and are recommended by clinical guidelines in the UK, United States, and Europe [11,12]. Nitrofurantoin use was initially limited to those with an eGFR \geq 60 mL/minute/1.73 m², because of concerns about poorer efficacy in patients with lower eGFRs. In 2014, a review of the evidence [13] and a retrospective cohort study [14] prompted the UK Medicines and Healthcare products Regulation Authority to lower the threshold for nitrofurantoin use to an eGFR \geq 45 mL/minute/1.73 m². However, outcomes following empirical nitrofurantoin prescribing in older adults with a UTI and an eGFR <60 mL/minute/1.73 m² are vet to be fully evaluated. There are also concerns about trimethoprim use in older adults with renal impairment, with increasing evidence of an association with hyperkalaemia and sudden death, especially when prescribed to patients already taking angiotensin-converting enzyme inhibitors, angiotensin-II receptor blockers, or potassium-sparing diuretics [15-19].

We used data from anonymised linked health records to estimate the risk of adverse outcomes in older patients with renal impairment empirically treated for suspected UTI in primary care. We firstly compared outcomes by eGFR to understand whether severity of renal impairment was associated with risk of adverse outcome following a UTI. This would help identify which patients would most benefit from interventions that improved prevention and/ or treatment of UTI. We also compared outcomes for older patients with an eGFR <60 mL/ minute/1.73 m² who were prescribed empirical nitrofurantoin versus empirical trimethoprim to inform prescribing decisions and explore if nitrofurantoin prescribing is safe in patients with renal impairment.

Methods

Data source

We used the Clinical Practice Research Datalink (CPRD), an electronic database of anonymised primary care records covering 11.3 million patients from 674 general practices across the UK [20]. Approximately 7% of the UK population are included and patients are broadly representative of the wider UK population in terms of age, sex, and ethnicity. The CPRD holds data on demographics, clinical encounters, diagnoses (coded using Read codes), drug prescriptions, laboratory tests, and referrals to specialists. Data are available once they have met a series of quality checks on completeness and reliability and the CPRD deems them to be of the standard required for research purposes. Linked hospital and death registration data are available for patients from approximately 50% of contributing English practices. Hospital diagnoses and causes of death are recorded using the International Classification of Diseases-10th Revision (ICD-10).

The CPRD Independent Scientific Advisory Committee approved the study protocol and analysis plan (protocol number 17_250, <u>S1 Protocol</u>). Further ethical approval was not required as the proposed research was within the remit of the CPRD's broad National Research Ethics Service approval. We used the Reporting of Studies Conducted using Observational Routinely-collected Health Data (RECORD) statement and checklist (S1 RECORD Checklist) to guide study reporting [21].

Design and participants

This was a retrospective cohort study using linked health record data. Patients were eligible for inclusion if, between 1 January 2010 and 31 December 2016, their data were of the quality required by CPRD, they were >65 years old, and eligible for data linkage. Only patients registered with practices that had consented to data linkage would have linked hospital and death registry data. We excluded patients if they were temporary residents (i.e., they registered with the practice for an acute problem but this was not their normal 'long-term' practice and thus medical record data would be limited) or had periods during their registration with the practice for which CPRD was unable to collect data, potentially leading to incomplete exposure/ event capture. We identified eligible patients with a Read code indicating an incident primary care presentation with a suspected UTI (codes available in S1 Appendix), a prescription code indicating same-day empirical prescribing of a relevant antibiotic, and a creatinine record in the preceding 24 months. We defined 'incident' as a presentation without a previous consultation with a UTI-related Read code, or trimethoprim or nitrofurantoin prescription in the preceding 90 days. We used the first incident episode during each patient's follow-up period. We excluded UTI episodes with a hospital discharge in the preceding 14 days to exclude hospitalacquired infections.

Exposures

We used the most recent serum creatinine value recorded in the 24 months preceding the incident UTI and data for patient age, gender, and ethnicity to calculate an eGFR as per the Modification of Diet in Renal Disease (MDRD) Study equation [22]. We categorised eGFRs as \geq 60 mL/minute/1.73 m², 45–59 mL/minute/1.73 m², 30–44 mL/minute/1.73 m², 15–29 mL/minute/1.73 m², and <15 mL/minute/1.73 m². These categories are similar to those used by the UK National Institute of Health and Care Excellence to categorise the stages of chronic kidney disease, except we combined eGFRs of 60–89 mL/minute/1.73 m² with those \geq 90 mL/minute/1.73 m². This was because eGFRs \geq 60 mL/minute/1.73 m² without additional evidence of kidney damage are clinically regarded as normal, and previous research found no difference in infection incidence or outcome between these two groups [5]. We used those with an eGFR \geq 60 mL/minute/1.73 m² as the reference and compared rates of adverse outcomes against the four other eGFR categories. We used the recorded empirical antibiotic prescription as the exposure variable to compare risk of adverse outcomes between patients with eGFRs <60 mL/minute/1.73 m² prescribed trimethoprim versus nitrofurantoin.

Outcomes

We assessed the impact of our stated exposures on the following adverse outcomes for patients empirically treated in primary care for an incident suspected UTI:

- Reconsultation for urinary symptoms and a same-day antibiotic prescription within 14 days following the incident UTI, as a proxy for treatment nonresponse, ascertained through Read and prescription codes recorded in primary care records.
- Hospitalisation for UTI, sepsis, or acute kidney injury (AKI) within 14 days following the incident UTI ascertained from ICD-10 codes recorded in linked hospital admission data for the first episode of a hospital admission, i.e., the episode most likely responsible for the admission.
- 3. Death within 28 days following the incident UTI using linked death registration data.

We also initially planned to include hospitalisation for pyelonephritis as an outcome. However, our exploratory work showed that pyelonephritis was rarely coded in hospital records (only 8 events in total) and thus was unlikely to be a reliable outcome for use with these data.

Statistical analyses

We used primary care demographic and clinical codes to describe baseline characteristics for patients by exposure status. To assess the impact of eGFR, we compared rates of each outcome in those with an eGFR \geq 60 mL/minute/1.73 m² to those in each category related to an eGFR <60 mL/minute/1.73 m², and used logistic regression to estimate odds ratios (ORs) and 95% confidence intervals (CIs). We adjusted for potential confounders of the association between renal impairment and outcome, including age; Index of Multiple Deprivation score quintile; Charlson comorbidity score [23]; the presence or absence of a record indicating diabetes, dementia, coronary heart disease, stroke, cancer, and heart failure; and polypharmacy. We inferred the presence of polypharmacy if the patient's record showed repeated monthly prescribing of \geq 5 medications in the year prior to the incident UTI. We also adjusted for the choice and duration of antibiotic therapy used to treat the incident UTI.

To assess the impact of empirical trimethoprim versus nitrofurantoin prescribing, we used a range of demographic and clinical variables to match patients on their propensity to receive a trimethoprim prescription. These included the confounders listed above and presence or absence of a record indicating urinary incontinence or long-term catheterisation, and longterm prescribing of angiotensin-converting enzyme inhibitors, angiotensin-II receptor blockers, or potassium-sparing diuretics. We used nearest neighbour matching with no replacement and matched each patient with a nitrofurantoin prescription to three patients with trimethoprim prescriptions. We assessed balance in measured baseline covariates between matched groups by visually inspecting jitter plots and histograms of covariate distribution before and after matching, and by calculating standardised mean differences for covariates between groups. We regarded standardised mean differences of <0.1 as reflecting adequate balance [24,25]. We used mixed effects logistic regression to account for clustering by general practice and calculated ORs and 95% CIs for each outcome.

All statistical tests were two-sided, with p < 0.05 considered statistically significant but an effect size of 10% considered clinically significant. *p*-values were derived using two-tailed Wald tests. Analyses were conducted in R version 3.2.1.

Results

From a cohort of 795,484 patients aged 65 and over, we identified 123,607 with an incident UTI empirically treated with a relevant antibiotic (Fig 1). Of these, 116,945 (95%) patients had

795,484 patients \geq 65 years old between 1st January 2010 and 31st December 2016 with data of the required quality and eligible for data linkage.



Fig 1. Flow of patients from initial identification in the database through to final cohort. UTI, urinary tract infection.

https://doi.org/10.1371/journal.pmed.1002652.g001

a creatinine measurement recorded in the 24 months prior to the incident UTI. In this final cohort, 32,428 (28%) were male and the median age at the time of incident UTI was 76 years (IQR 70–83). Almost one third of creatinine measurements were in the 90 days prior to the incident UTI. Median duration between most recent creatinine and UTI was 169 days (IQR 65–285). Using the MDRD study equation, 76,112 (65.1%) of patients were assigned an eGFR \geq 60, 26,970 (23.1%) an eGFR of 45–59, 10,854 (9.3%) an eGFR of 30–44, 2,667 (2.3%) an eGFR of 15–29, and 342 (0.3%) an eGFR of <15. Baseline characteristics showed that patients with lower eGFRs had a relatively greater number of comorbidities and comprised greater proportions of patients with polypharmacy (Table 1). Trimethoprim was the most commonly prescribed empirical antibiotic across all eGFR groups. Nitrofurantoin was the second most common except in patients with an eGFR <15 mL/minute/1.73 m².

Outcomes according to calculated eGFR

There were 7,203 reconsultations with urinary symptoms resulting in another antibiotic prescription within 14 days of the incident UTIs, equating to about 6% of the cohort. The odds of reconsulting and receiving another antibiotic prescription were no different between those with an eGFR \geq 60 mL/minute/1.73 m² and those with eGFRs <60mL/minute/1.73 m² (Table 2).

There were 1,991 hospitalisations for UTI (1.7% of the cohort), 176 for sepsis (0.2% of the cohort), and 865 for AKI (0.7% of the cohort) within 14 days of the incident UTIs. Compared to those with an eGFR \geq 60 mL/minute/1.73 m², odds of hospitalisation for UTI increased in those with eGFRs of 45–59 (adjusted OR 1.14, 95% CI 1.01–1.28), 30–44 (adjusted OR 1.25, 95% CI 1.08–1.44), 15–29 (adjusted OR 1.76, 95% CI 1.43–2.16), and <15 (adjusted OR 1.68, 95% CI 1.01–2.82). Odds of hospitalisation for sepsis were no different in those with an eGFR

Table 1. Baseline characteristics of included patients according to eGFR category.

Characteristics	eGFR ≥60	eGFR 45-59	eGFR 30-44	eGFR 15-29	eGFR <15
N	76,112 (65.1)	26,970 (23.1)	10,854 (9.3)	2,667 (2.3)	342 (0.3)
Men	21,816 (28.7)	6,674 (24.7)	2,894 (26.7)	880 (33.0)	164 (48.0)
Mean (SD) age	75.2 (7.9)	79.1 (8.3)	82.5 (8.0)	83.2 (8.0)	79 (8.2)
Prescribed antibiotic					
Amoxicillin	3,370 (4.4)	1,367 (5.1)	680 (6.3)	289 (10.8)	45 (13.2)
Cefalexin	4,168 (5.5)	1,749 (6.5)	917 (8.4)	319 (12.0)	47 (13.7)
Ciprofloxacin	2,344 (3.1)	811 (3.0)	388 (3.6)	148 (5.5)	26 (7.6)
Co-amoxiclav	3,170 (4.2)	1,227 (4.5)	613 (5.6)	208 (7.8)	44 (12.9)
Nitrofurantoin	16,719 (22.0)	5,237 (19.4)	1,815 (16.7)	391 (14.7)	41 (12.0)
Trimethoprim	46,341 (60.9)	16,579 (61.5)	6,441 (59.3)	1,312 (49.2)	139 (40.6)
Index of multiple deprivation decile					
1 or 2 (least deprived)	19,939 (26.2)	6,401 (23.7)	2,292 (21.1)	560 (21.0)	65 (19.0)
3 or 4	18,413 (24.2)	6,524 (24.2)	2,526 (23.3)	619 (23.2)	77 (22.5)
5 or 6	16,606 (21.8)	5,859 (21.7)	2,443 (22.5)	615 (23.1)	85 (24.9)
7 or 8	12,283 (16.1)	4,626 (17.2)	1,918 (17.7)	477 (17.9)	45 (13.2)
9 or 10 (most deprived)	8,8/1 (11./)	3,560 (13.2)	1,6/5 (15.4)	396 (14.8)	70 (20.5)
Housebound	2,261 (3.0)	1,201 (4.5)	866 (8.0)	267 (10.0)	34 (9.9)
Respiratory disease	15,853 (20.8)	5,521 (20.5)	2,208 (20.3)	538 (20.2)	46 (13.5)
Cardiac failure	2,187 (2.9)	1,733 (6.4)	1,367 (12.6)	499 (18.7)	56 (16.4)
Dementia	3,724 (4.9)	2,024 (7.5)	1,129 (10.4)	265 (9.9)	21 (6.1)
Peripheral vascular disease	2,903 (3.8)	1,502 (5.6)	953 (8.8)	319 (12.0)	43 (12.6)
Rheumatoid arthritis	2,140 (2.8)	803 (3.0)	339 (3.1)	97 (3.6)	9 (2.6)
Cancer	11,291 (14.8)	4,211 (15.6)	1,891 (17.4)	498 (18.7)	82 (24.0)
Stroke	6,714 (8.8)	3,123 (11.6)	1,695 (15.6)	454 (17.0)	67 (19.6)
Diabetes	11,956 (15.7)	5,103 (18.9)	2,863 (26.4)	961 (36.0)	112 (32.7)
Liver disease	445 (0.6)	165 (0.6)	69 (0.6)	11 (0.4)	3 (0.9)
Ischaemic heart disease	11,611 (15.3)	5,814 (21.6)	3,118 (28.7)	878 (32.9)	114 (33.3)
Urinary catheter	2,360 (3.1)	854 (3.2)	504 (4.6)	193 (7.2)	39 (11.4)
Urinary incontinence	10,966 (14.4)	4,089 (15.2)	1,702 (15.7)	398 (14.9)	41 (12.0)
Polypharmacy	24,478 (32.2)	11,419 (42.3)	6,371 (58.7)	1,797 (67.4)	237 (69.3)
Potassium-sparing diuretic	1,470 (1.9)	946 (3.5)	732 (6.7)	158 (5.9)	4 (1.2)
Angiotensin-converting enzyme inhibitor	16,430 (21.6)	7,586 (28.1)	3,446 (31.7)	718 (26.9)	46 (13.5)
Angiotensin-II receptor antagonist	8,195 (10.8)	3,933 (14.6)	1,885 (17.4)	453 (17.0)	42 (12.3)
Charlson score					
0	30,663 (40.3)	6,131 (22.7)	879 (8.1)	104 (3.9)	6 (1.8)
1	18,770 (24.7)	4,100 (15.2)	824 (7.6)	116 (4.3)	12 (3.5)
2	12,973 (17.0)	6,309 (23.4)	2,621 (24.1)	504 (18.9)	81 (23.7)
3	7,394 (9.7)	4,666 (17.3)	2,424 (22.3)	577 (21.6)	64 (18.7)
4	3,219 (4.2)	2,725 (10.1)	1,728 (15.9)	481 (18.0)	66 (19.3)
5	1,636 (2.1)	1,661 (6.2)	1,148 (10.6)	401 (15.0)	45 (13.2)
<u>≥</u> 6	1,457 (1.9)	1,378 (5.1)	1,230 (11.3)	484 (18.1)	68 (19.9)

Numbers are values (%) unless otherwise stated.

Abbreviation: eGFR, estimated glomerular filtration rate.

https://doi.org/10.1371/journal.pmed.1002652.t001

of 45–59 but were significantly higher in those with eGFRs of 30–44 (adjusted OR 1.70, 95% CI 1.06–2.72), 15–29 (adjusted OR 2.72, 95% CI 1.50–4.94), and <15 (adjusted OR 4.24, 95% CI 1.48–11.23). The risk of hospitalisation for AKI increased in a graded manner relative to renal function, with adjusted ORs of 1.57 (95% CI 1.29–1.91), 3.21 (95% CI 2.61–3.94), 6.70 (95% CI 5.24–8.55), and 4.53 (95% CI 2.52–8.17) for eGFRs of 45–59, 30–44, 15–29, and <15 mL/minute/1.73 m², respectively.

Table 2. Adjusted ORs and 95% CIs for each outcome by eGFR category.

Reconsultation and re-prescription within 14 days	Number of UTIs	Number (%) of events	Crude OR	Adjusted [*] OR (95% CI)	<i>p</i> -value
eGFR ≥60	76,112	4,852 (6.4)	1	1	
eGFR 45–59	26,970	1,563 (5.8)	0.90	0.97 (0.91-1.03)	0.328
eGFR 30-44	10,854	626 (5.8)	0.90	1.03 (0.94–1.14)	0.511
eGFR 15–29	2,667	148 (5.5)	0.86	1.02 (0.86-1.22)	0.804
eGFR <15	342	14 (4.1)	0.63	0.72 (0.42-1.23)	0.224
Hospitalised for UTI within 14 days					
$eGFR \ge 60$	76,112	1,003 (1.3)	1	1	
eGFR 45-59	26,970	526 (2.0)	1.49	1.14 (1.01–1.28)	0.028
eGFR 30-44	10,854	317 (2.9)	2.25	1.25 (1.08–1.44)	0.003
eGFR 15–29	2,667	129 (4.8)	3.81	1.76 (1.43-2.16)	< 0.001
eGFR <15	342	16 (4.7)	3.68	1.68 (1.01-2.82)	< 0.001
Hospitalised for sepsis within 14 days					
$eGFR \ge 60$	76,112	77 (0.1)	1	1	
eGFR 45-59	26,970	46 (0.2)	1.69	1.36 (0.92-2.01)	0.119
eGFR 30-44	10,854	32 (0.3)	2.92	1.70 (1.06-2.72)	0.027
eGFR 15–29	2,667	17 (0.6)	6.33	2.72 (1.50-4.94)	< 0.001
eGFR <15	342	4 (1.2)	11.69	4.24 (1.48-11.23)	0.007
Hospitalised for AKI within 14 days					
$eGFR \ge 60$	76,112	280 (0.4)	1	1	
eGFR 45-59	26,970	204 (0.8)	2.06	1.57 (1.29–1.91)	< 0.001
eGFR 30-44	10,854	231 (2.1)	5.89	3.21 (2.61-3.94)	< 0.001
eGFR 15–29	2,667	137 (5.1)	14.67	6.70 (5.24-8.55)	< 0.001
eGFR <15	342	13 (3.8)	10.70	4.53 (2.52-8.17)	< 0.001
Death within 28 days					
eGFR ≥60	76,112	588 (0.8)	1	1	
eGFR 45-59	26,970	285 (1.1)	1.37	0.92 (0.79–1.07)	0.275
eGFR 30-44	10,854	201 (1.9)	2.42	1.05 (0.87–1.26)	0.598
eGFR 15-29	2,667	99 (3.7)	4.95	1.63 (1.27–2.10)	< 0.001
eGFR <15	342	19 (5.6)	7.56	2.37 (1.44–3.89)	< 0.001

*ORs adjusted for age, gender, Index of Multiple Deprivation score quintile, Charlson comorbidity score, being housebound, respiratory disease, peripheral vascular disease, liver disease, rheumatoid arthritis, diabetes, dementia, coronary heart disease, stroke, cancer, heart failure, polypharmacy, long-term prescribing of angiotensin-converting enzyme inhibitors, angiotensin-II receptor blockers, or potassium-sparing diuretics, urinary catheter, urinary incontinence, and choice and duration of antibiotic therapy.

Abbreviations: AKI, acute kidney injury; CI, confidence interval; eGFR, estimated glomerular filtration rate; OR, odds ratio; UTI, urinary tract infection.

https://doi.org/10.1371/journal.pmed.1002652.t002

There were 1,162 deaths in the 28 days following the incident UTIs, equating to about 1% of the cohort. Compared to those with an eGFR \geq 60 mL/minute/1.73 m², the odds of death were no different in those with an eGFR \geq 30 mL/minute/1.73 m², 63% higher in those with an eGFR of 15–29 (adjusted OR 1.63, 95% CI 1.27–2.10), and over 2-fold higher in those with an eGFR <15 (adjusted OR 2.37, 95% CI 1.44–3.89).

Trimethoprim versus nitrofurantoin in those with an eGFR ${<}60$ mL/ minute/1.73 m^2

Of the 40,833 patients with an eGFR <60 mL/minute/1.73 m², 24,471 (60%) were prescribed trimethoprim and 7,484 (18%) were prescribed nitrofurantoin. We matched 20,948 patients with an eGFR of 45–60 (15,711 prescribed trimethoprim, 5,237 prescribed nitrofurantoin),

7,260 with an eGFR of 30-44 (5,445 prescribed trimethoprim, 1,815 prescribed nitrofurantoin), and 1,728 with an eGFR <30 (1,296 prescribed trimethoprim, 432 prescribed nitrofurantoin). Inspection of jitter plots and histograms suggested matching had improved balance of covariates across trimethoprim versus nitrofurantoin groups. Standardised mean differences were all less than 0.1 (Table 3).

Empirical nitrofurantoin prescribing was associated with lower odds of hospitalisation for AKI across all eGFR groups (eGFR 45–59: OR 0.62, 95% CI 0.40–0.94; eGFR 30–44: OR 0.47, 95% CI 0.30–0.73; eGFR <30: OR 0.45, 95% CI 0.25–0.81) (Table 4). Nitrofurantoin was also associated with lower odds of reconsultation and re-prescription in patients with eGFRs of 45–59 (OR 0.74, 95% CI 0.61–0.91) and lower odds of death in patients with eGFRs of 30–44 (OR 0.61, 95% CI 0.39–0.95). There were no other statistically significant differences between empirical trimethoprim versus nitrofurantoin prescribing. Importantly, we did not detect any increase in odds of adverse outcomes in patients prescribed nitrofurantoin.

Sensitivity analyses

We restricted our eGFR and outcomes analysis to patients with a creatinine measured in the 90 days prior to the incident UTI, to increase the likelihood that the calculated eGFR reflected their current renal function. Results were consistent with our main analysis and most of the statistically significant ORs increased in magnitude (S1 Table). We also combined the hospitalisation and death outcomes in our trimethoprim versus nitrofurantoin analysis to increase statistical power to detect these adverse outcomes (S2 Table). Findings were consistent with our main analysis.

Discussion

Our results show that compared to an eGFR of >60 mL/minute/1.73 m², older patients with an eGFR of <60 mL/minute/1.73 m² who were empirically treated for suspected UTI in primary care had greater odds of hospitalisation for UTI and AKI, those with an eGFR <45 had greater odds of hospitalisation for sepsis, and those with an eGFR <30 had greater odds of death. The magnitude of each association generally increased relative to the severity of the renal impairment. We also showed that, compared to trimethoprim, nitrofurantoin was associated with reduced odds of hospitalisation for AKI across all eGFR groups and was not associated with an increased risk of any adverse event evaluated in our study.

Results in context

Previous research focussed on the risk of infection-related hospitalisation in adults with renal impairment and showed a greater risk of hospitalisation for pneumonia, UTI, bacteraemia, and cellulitis in those with eGFRs <60 mL/minute/1.73 m² [4–6]. Previous studies also showed a greater risk of death following an infection-related hospitalisation in patients with renal impairment [3–5] but provided little information on health service contact prior to the adverse outcome, thus limiting interpretation about possible opportunities to intervene. Our study shows an increased risk of infection-related hospitalisation and death in older adults with renal impairment, following infection-related presentation and treatment in primary care. We also show increased odds of AKI hospitalisation in those with lower initial eGFRs, previously only investigated in a small cohort (n = 790) of patients hospitalised with UTI, who likely had more severe infection on initial presentation [26]. To the best of our knowledge, this is the first study to investigate the impact of eGFR on odds of reconsultation and re-prescription following an infection-related illness. We found no difference in the odds of this outcome across the different eGFR groups, suggesting that the increased odds of UTI, sepsis, and death were less

Table 3.	Balance of baseline characteristics across trimethoprim and nitrofurantoin groups following propensity score matching for patients with renal
impairm	nent.

eGFR 45-60 mL/minute/1.73 m ²	Trimethoprim	Nitrofurantoin	Standardised mean difference
N	15,711	5,237	
Men	3,421 (21.8)	1,120 (21.4)	0.01
Mean (SD) age	78.9 (8.3)	78.8 (8.2)	-0.02
Index of multiple deprivation decile			
1 or 2 (least deprived)	3,746 (23.8)	1,305 (24.9)	
3 or 4	3,807 (24.2)	1,257 (24.0)	0.00
5 or 6	3,488 (22.2)	1,088 (20.8)	
/ or 8 9 or 10 (most deprived)	2,711 (17.3)	867 (16.6) 720 (13.7)	
Housebound	637 (4.1)	206 (3.9)	-0.01
Respiratory disease	3,173 (20,2)	1,115 (21.3)	0.00
Cardiac failure	952 (6.1)	317 (6.1)	0.00
Dementia	1,111 (7,1)	361 (6.9)	-0.01
Cancer	2 389 (15 2)	853 (16 3)	0.03
Stroke	1 768 (11 3)	611 (11 7)	0.01
Diabetes	2 890 (18.4)	956 (18.3)	0.00
Instruction boost disease	2,000 (10.4)	1,008 (21)	0.00
Ischaemic neart disease	3,244 (20.0)	1,090 (21)	0.01
	406 (2.6)	1/0 (3.4)	0.00
	2,337 (14.9)	883 (16.9)	0.05
Polypharmacy	6,497 (41.4)	2,262 (43.2)	0.04
Potassium-sparing diuretic	551 (3.5)	197 (3.8)	0.01
Angiotensin-converting enzyme inhibitor	4,364 (27.8)	1,437 (27.4)	-0.01
Angiotensin-II receptor antagonist	2,294 (14.6)	786 (15)	0.01
Charlson score			
0	3,802 (24.2)	1,212 (23.1)	
1	2,426 (15.4)	814 (15.5)	0.02
3	2,632 (16.8)	904 (17.3)	
4	1,561 (9.9)	490 (9.4)	
5	891 (5.7)	313 (6.0)	
<u>≥6</u>	730 (4.6)	248 (4.7)	
eGFR 30-44 mL/minute/1.73 m ²	Trimethoprim	Nitrofurantoin	Standardised mean difference
<u>N</u>	5,445	1,815	
Men	1,201 (22.1)	414 (22.8)	-0.02
Mean (SD) age	82.3 (7.9)	82.2 (8.0)	-0.01
Index of multiple deprivation decile			
1 or 2 (least deprived)	1,162 (21.3)	360 (19.8)	
3 or 4	1,219 (22.4)	466 (25.7)	0.00
7 or 8	986 (18.1)	394 (21.7) 301 (16.6)	
9 or 10 (most deprived)	849 (15.6)	294 (16.2)	
Housebound	475 (8.7)	167 (9.2)	0.02
Respiratory disease	1,148 (21.1)	384 (21.2)	0.00
Cardiac failure	657 (12.1)	226 (12.5)	0.01
Dementia	605 (11.1)	201 (11.1)	0.00
Cancer	924 (17)	314 (17.3)	0.01
Stroke	880 (16.2)	304 (16.7)	0.02
Diabetes	1,493 (27.4)	527 (29)	0.04
Ischaemic heart disease	1,517 (27.9)	503 (27.7)	0.01
Urinary catheter	240 (4.4)	112 (6.2)	0.07
Urinary incontinence	882 (16 2)	302 (16.6)	0.07
	002 (10.2)	302 (10.0)	0.01

(Continued)

Table 3. (Continued)

Polypharmacy	3,302 (60.6)	1,117 (61.5)	0.02
Potassium-sparing diuretic	391 (7.2)	137 (7.5)	0.01
Angiotensin-converting enzyme inhibitor	1,758 (32.3)	593 (32.7)	0.01
Angiotensin-II receptor antagonist	941 (17.3)	307 (16.9)	-0.01
Charlson score			
0 1 2 3 4 5	441 (8.1) 441 (8.1) 1,283 (23.6) 1,237 (22.7) 894 (16.4) 550 (10 1)	147 (8.1) 144 (7.9) 442 (24.4) 381 (21.0) 273 (15.0) 213 (11 7)	0.02
<u>≥</u> 6	599 (11.0)	215 (11.8)	
eGFR <30 mL/minute/1.73 m ²	Trimethoprim	Nitrofurantoin	Standardised mean difference
N	1,296	432	
Men	339 (26.2)	113 (26.2)	0.00
Mean (SD) age	83.4 (8.2)	83.4 (8.0)	0.00
Index of multiple deprivation decile			
1 or 2 (least deprived) 3 or 4 5 or 6 7 or 8 9 or 10 (most deprived)	281 (21.7) 297 (22.9) 286 (22.1) 231 (17.8) 201 (15.5)	93 (21.5) 95 (22.0) 100 (23.1) 79 (18.3) 65 (15.0)	0.02
Housebound	131 (10.1)	42 (9.7)	-0.01
Respiratory disease	250 (19.3)	86 (19.9)	0.02
Cardiac failure	224 (17.3)	71 (16.4)	-0.02
Dementia	132 (10.2)	42 (9.7)	-0.02
Cancer	229 (17.7)	74 (17.1)	-0.01
Stroke	215 (16.6)	71 (16.4)	0.00
Diabetes	451 (34.8)	155 (35.9)	0.02
Ischaemic heart disease	406 (31.3)	132 (30.6)	-0.02
Urinary catheter	93 (7.2)	29 (6.7)	-0.02
Urinary incontinence	193 (14.9)	63 (14.6)	-0.01
Polypharmacy	877 (67.7)	294 (68.1)	0.01
Potassium-sparing diuretic	76 (5.9)	23 (5.3)	-0.02
Angiotensin-converting enzyme inhibitor	340 (26.2)	112 (25.9)	-0.01
Angiotensin-II receptor antagonist	220 (17.0)	79 (18.3)	0.03
Charlson score			
$ \begin{array}{c} 0\\ 1\\ 2\\ 3\\ 4\\ 5\\ \geq 6 \end{array} $	51 (3.9) 57 (4.4) 291 (22.5) 276 (21.3) 229 (17.7) 179 (13.8) 213 (16.4)	21 (4.9) 21 (4.9) 92 (21.3) 93 (21.5) 79 (18.3) 56 (13.0) 70 (16.2)	-0.03

Numbers are values (%) unless otherwise stated. Abbreviation: eGFR, estimated glomerular filtration rate.

https://doi.org/10.1371/journal.pmed.1002652.t003

likely to be due to treatment nonresponse and more likely to be related to other patient or renal factors.

Trimethoprim (with or without sulfamethoxazole) prescribing is associated with an increased risk of hyperkalaemia, AKI, and death, compared to amoxicillin [16–19].

eGFR 45-59	Trimethoprim group, $n = 15,711$	Nitrofurantoin group, $n = 5,237$	OR (95% CI)	<i>p</i> -value
	Number (%) of events	Number (%) of events		
Death within 28 days	159 (1.0)	50 (1.0)	0.94 (0.69–1.30)	0.718
Reconsultation and re-prescription within 14 days	942 (6.0)	290 (5.5)	0.74 (0.61-0.91)	0.004
Hospitalised for UTI within 14 days	288 (1.8)	105 (2.0)	1.09 (0.74-1.61)	0.648
Hospitalised for sepsis within 14 days	25 (0.2)	6 (0.72)	0.72 (0.30-1.76)	0.470
Hospitalised for AKI within 14 days	126 (0.8)	26 (0.5)	0.62 (0.40-0.94)	0.025
eGFR 30-44	Trimethoprim group, $n = 5,445$	Nitrofurantoin group, $n = 1,815$		
	Number (%) of events	Number (%) of events	OR (95% CI)	<i>p</i> -value
Death within 28 days	113 (2.1)	23 (1.3)	0.61 (0.39-0.95)	< 0.001
Reconsultation and re-prescription within 14 days	318 (5.8)	117 (6.4)	0.98 (0.71-1.33)	0.874
Hospitalised for UTI within 14 days	168 (3.1)	57 (3.1)	0.80 (0.44-1.47)	0.482
Hospitalised for sepsis within 14 days	14 (0.3)	2 (0.1)	0.43 (0.10-1.88)	0.262
Hospitalised for AKI within 14 days	146 (2.7)	23 (1.3)	0.47 (0.30-0.73)	< 0.001
eGFR <30	Trimethoprim group, $n = 1,296$	Nitrofurantoin group, $n = 432$		
	Number (%) of events	Number (%) of events	OR (95% CI)	<i>p</i> -value
Death within 28 days	49 (3.8)	18 (4.2)	1.11 (0.64–1.93)	0.713
Reconsultation and re-prescription within 14 days	74 (5.7)	29 (6.7)	1.19 (0.76-1.85)	0.446
Hospitalised for UTI within 14 days	73 (5.6)	23 (5.3)	0.94 (0.58-1.53)	0.808
Hospitalised for sepsis within 14 days	8 (0.6)	2 (0.5)	0.75 (0.16-3.54)	0.715
Hospitalised for AKI within 14 days	84 (6.5)	13 (3.0)	0.45 (0.25-0.81)	0.008

Table 4. ORs and 95% CIs for each outcome in propensity-score* matched trimethoprim versus nitrofurantoin groups, across three eGFR categories.

*The following baseline variables were used in the propensity-score model: age, gender, Index of Multiple Deprivation score quintile, Charlson comorbidity score, being housebound, the presence or absence of respiratory disease, peripheral vascular disease, liver disease, rheumatoid arthritis, diabetes, dementia, coronary heart disease, stroke, cancer, heart failure, urinary catheter, urinary incontinence, polypharmacy, long-term prescribing of angiotensin-converting enzyme inhibitors, angiotensin-II receptor blockers, or potassium-sparing diuretics.

Abbreviations: AKI, acute kidney injury; CI, confidence interval; eGFR, estimated glomerular filtration rate; OR, odds ratio; UTI, urinary tract infection.

https://doi.org/10.1371/journal.pmed.1002652.t004

Amoxicillin accounts for only about 5% of prescribing for UTI in the UK [9] and thus is a less helpful comparator for clinical decision-making. Furthermore, these studies did not investigate associations by degree of renal impairment, providing little information to guide prescribing in this population. Two studies assessed trimethoprim and nitrofurantoin prescribing in patients with renal impairment. The first compared treatment failure rates in women with UTI prescribed nitrofurantoin according to renal function and found no difference across the eGFR groups [14]. This study lacked a comparator group prescribed an alternative antibiotic, which makes it difficult to interpret their findings. The second compared outcomes in older women with a median eGFR of 38 mL/minute/1.73 m², prescribed either nitrofurantoin or trimethoprim, and found no difference in risk of treatment failure or UTI hospitalisation [27]. We compared nitrofurantoin with trimethoprim across three eGFR groups and found that nitrofurantoin was associated with lower odds of reconsultation and re-prescription in patients with eGFRs of 45–59. This could be explained by recent data showing that 34% of communityacquired Escherichia coli UTIs in England are resistant to trimethoprim, compared to only 2.7% resistant to nitrofurantoin [28]. We did not find statistically significant differences between reconsultation and re-prescription rates in people with eGFRs < 45. This could be due to less statistical power, as nitrofurantoin use was less common in these patients because of the advice to use with care in patients with eGFRs of 30–44 and to avoid in eGFRs <30. It may also be due to the possibility that nitrofurantoin efficacy was reduced in those with lower eGFRs but was offset by the high rates of trimethoprim resistance and thus resulted in

apparent similar rates of reconsultation and re-prescription. Our finding that nitrofurantoin was associated with a reduced risk of death in those with moderate renal impairment is consistent with previously reported estimates in studies that compared nitrofurantoin with amoxicillin in the general population [18,19]. We also found a previously unreported lower risk of AKI associated with nitrofurantoin use across all three eGFR groups of our cohort.

Strengths and weaknesses of this study

We used data from a general practice database that is broadly representative of the UK population, increasing the generalisability of our findings. This is the largest cohort study to investigate the impact of eGFR on infection-related outcomes, with a sample size >4 times larger than the previously largest study [4]. Cohort entry was dependent on presentation and empirical treatment of UTI in primary care, and thus reduced indication bias. We adjusted for the presence/absence of more comorbidities than previous studies, increasing the likelihood of an independent association between eGFR and adverse outcomes. This is the first study to investigate trimethoprim versus nitrofurantoin prescribing in renal impairment, using clinically relevant eGFR groups analogous to stages of chronic kidney disease and without excluding men. We also reduced indication bias by matching patients on their propensity to receive trimethoprim, and achieving adequate balance of covariates across the two groups.

Our study has important limitations. We attempted to capture patients presenting with UTI but had no microbiological data to support this. However, whilst a limitation, this is also more representative of clinical practice. We were unable to investigate pulmonary/hepatic toxicity related to nitrofurantoin use because of the lack of reliable codes, and differential use of these codes by clinicians. However, two systematic reviews have shown that these toxicities are rare with short-term use [29,30]. We relied on a creatinine measurement from the 24 months prior to the UTI to estimate an eGFR, but this may not fully represent patients' current renal function. Finally, despite our design, differential coding, indication bias, and residual confounding may still have affected our findings.

Clinical implications

Around 20% of adults aged \geq 65 present to primary care at least once with a UTI [9], and around 20% have renal impairment [2] and thus are at greater risk of an adverse outcome. The initial primary care visit presents a potential opportunity to address this increased risk. However, recommended interventions such as stopping co-prescribed angiotensin-converting enzyme inhibitors or angiotensin-II receptor blockers have not been evaluated in the primary care setting [31]. Therefore, there is a need for research that evaluates primary care-based interventions that may prevent adverse outcomes, including AKI, in patients with renal impairment and community-acquired infection. The absolute risks of hospitalisation and death are low. Therefore, interventions may be best targeted at those at highest risk—i.e., those with more severe renal impairment.

Current guidelines and the British National Formulary limit nitrofurantoin use to those with an eGFR >45 mL/minute/1.73 m², although short courses can be used with care in those with eGFRs >30 mL/minute/1.73 m² [11]. We found no evidence to support this limitation and actually found nitrofurantoin to be associated with a reduced risk of AKI, compared to trimethoprim. Research suggesting poorer nitrofurantoin efficacy in patients with renal impairment assessed urinary nitrofurantoin excretion, not clinical outcomes, and was restricted to small samples [13]. Our findings, combined with increasing rates of bacterial resistance to trimethoprim and the importance of avoiding broader spectrum agents, support wider use of nitrofurantoin for older patients with low eGFRs.

Conclusion

Our findings show that older patients with renal impairment presenting to primary care with a UTI are at greater risk of adverse outcomes independent of other comorbidities and of prescribed empirical antibiotic treatment. Despite documented concerns, we found no increased risk of adverse outcomes in patients with an eGFR <60 mL/minute/1.73 m² prescribed nitro-furantoin and support its wider use in selected patients with moderate-severe renal impairment.

Disclaimer: The views expressed in this publication are those of the authors and not necessarily those of the NIHR, NHS Wales, HCRW, or the Welsh Government.

Supporting information

S1 RECORD Checklist. RECORD statement. RECORD, reporting of studies conducted using observational routinely-collected health data. (DOCX)

S1 Appendix. Identifying clinically diagnosed UTI using read and ICD-10 codes. UTI, urinary tract infection. (DOCX)

S1 Table. Adjusted odds ratios and 95% confidence intervals for each outcome by eGFR category restricted to the 37,379 patients with a creatinine measurement in the 90 days prior to the UTI event. eGFR, estimated glomerular filtration rate. (DOCX)

S2 Table. Adjusted odds ratios and 95% confidence intervals for a combined 'hospitalisation or death' outcome in matched trimethoprim versus nitrofurantoin groups, across three eGFR categories. eGFR, estimated glomerular filtration rate. (DOCX)

S1 Protocol. Study protocol. (DOCX)

Author Contributions

Conceptualization: Haroon Ahmed, Daniel Farewell, Nick A. Francis, Christopher C. Butler.

Data curation: Haroon Ahmed.

Formal analysis: Haroon Ahmed, Daniel Farewell.

Funding acquisition: Haroon Ahmed, Daniel Farewell, Shantini Paranjothy, Christopher C. Butler.

Methodology: Haroon Ahmed, Daniel Farewell, Shantini Paranjothy, Christopher C. Butler.

Project administration: Haroon Ahmed.

Supervision: Daniel Farewell, Nick A. Francis, Shantini Paranjothy, Christopher C. Butler.

Writing – original draft: Haroon Ahmed.

Writing – review & editing: Daniel Farewell, Nick A. Francis, Shantini Paranjothy, Christopher C. Butler.

References

- Inker LA, Astor BC, Fox CH, Isakova T, Lash JP, Peralta CA, et al. KDOQI US commentary on the 2012 KDIGO clinical practice guideline for the evaluation and management of CKD. Am J Kidney Dis. 2014; 63(5):713–35. Epub 2014/03/22. https://doi.org/10.1053/j.ajkd.2014.01.416 PMID: 24647050.
- 2. Chronic kidney disease prevalence model: Public Health England; 2014 [cited 2018 Jan 12]. Available from: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/612303/ ChronickidneydiseaseCKDprevalencemodelbriefing.pdf.
- 3. Wang HE, Gamboa C, Warnock DG, Muntner P. Chronic Kidney Disease and Risk of Death from Infection. Am J Nephrol. 2011; 34(4):330–6. https://doi.org/10.1159/000330673 PMID: 21860228.
- James MT, Laupland KB, Tonelli M, Manns BJ, Culleton BF, Hemmelgarn BR. Risk of bloodstream infection in patients with chronic kidney disease not treated with dialysis. Arch Intern Med. 2008; 168 (21):2333–9. Epub 2008/11/26. https://doi.org/10.1001/archinte.168.21.2333 PMID: 19029498.
- Ishigami J, Grams ME, Chang AR, Carrero JJ, Coresh J, Matsushita K. CKD and Risk for Hospitalization With Infection: The Atherosclerosis Risk in Communities (ARIC) Study. Am J Kidney Dis. 2017; 69 (6):752–61. Epub 2016/11/26. https://doi.org/10.1053/j.ajkd.2016.09.018 PMID: 27884474; PubMed Central PMCID: PMC5438909.
- Dalrymple LS, Katz R, Kestenbaum B, de Boer IH, Fried L, Sarnak MJ, et al. The Risk of Infection-Related Hospitalization With Decreased Kidney Function. Am J Kidney Dis. 2012; 59(3):356–63. https://doi.org/10.1053/j.ajkd.2011.07.012 PMID: 21906862; PubMed Central PMCID: PMC3288732.
- Eleftheriadis T, Antoniadi G, Liakopoulos V, Kartsios C, Stefanidis I. Disturbances of acquired immunity in hemodialysis patients. Semin Dial. 2007; 20(5):440–51. Epub 2007/09/28. <u>https://doi.org/10.1111/j.</u> 1525-139X.2007.00283.x PMID: 17897251.
- Himmelfarb J. Uremic toxicity, oxidative stress, and hemodialysis as renal replacement therapy. Semin Dial. 2009; 22(6):636–43. Epub 2009/12/19. https://doi.org/10.1111/j.1525-139X.2009.00659.x PMID: 20017834.
- Ahmed H, Farewell D, Jones HM, Francis NA, Paranjothy S, Butler CC. Incidence and antibiotic prescribing for clinically diagnosed urinary tract infection in older adults in UK primary care, 2004–2014. PLoS ONE. 2018; 13(1):e0190521. Epub 2018/01/06. https://doi.org/10.1371/journal.pone.0190521 PMID: 29304056; PubMed Central PMCID: PMC5755802.
- Butler CC, Hawking MK, Quigley A, McNulty CA. Incidence, severity, help seeking, and management of uncomplicated urinary tract infection: a population-based survey. Br J Gen Pract. 2015; 65(639):e702– 7. Epub 2015/09/29. https://doi.org/10.3399/bjgp15X686965 PMID: 26412847; PubMed Central PMCID: PMC4582883.
- 11. SIGN. Management of suspected bacterial urinary tract infection in adults: Scottish Intercollegiate Guidelines Network; 2015 [cited 2018 March 30]. Available from: https://www.sign.ac.uk/sign-88-management-of-suspected-bacterial-urinary-tract-infection-in-adults.html
- Gupta K, Hooton TM, Naber KG, Wullt B, Colgan R, Miller LG, et al. International clinical practice guidelines for the treatment of acute uncomplicated cystitis and pyelonephritis in women: A 2010 update by the Infectious Diseases Society of America and the European Society for Microbiology and Infectious Diseases. Clin Infect Dis. 2011; 52(5):e103–20. Epub 2011/02/05. <u>https://doi.org/10.1093/cid/ciq257</u> PMID: 21292654.
- 13. Oplinger M, Andrews CO. Nitrofurantoin contraindication in patients with a creatinine clearance below 60 mL/min: looking for the evidence. Ann Pharmacother. 2013; 47(1):106–11. Epub 2013/01/24. https://doi.org/10.1345/aph.1R352 PMID: 23341159.
- Geerts AF, Eppenga WL, Heerdink R, Derijks HJ, Wensing MJ, Egberts TC, et al. Ineffectiveness and adverse events of nitrofurantoin in women with urinary tract infection and renal impairment in primary care. Eur J Clin Pharmacol. 2013; 69(9):1701–7. Epub 2013/05/11. https://doi.org/10.1007/s00228-013-1520-x PMID: 23660771
- Antoniou T, Gomes T, Juurlink DN, Loutfy MR, Glazier RH, Mamdani MM. Trimethoprim-sulfamethoxazole-induced hyperkalemia in patients receiving inhibitors of the renin-angiotensin system: a population-based study. Arch Intern Med. 2010; 170(12):1045–9. Epub 2010/06/30. <u>https://doi.org/10.1001/</u> archinternmed.2010.142 PMID: 20585070.
- Antoniou T, Gomes T, Mamdani MM, Yao Z, Hellings C, Garg AX, et al. Trimethoprim-sulfamethoxazole induced hyperkalaemia in elderly patients receiving spironolactone: nested case-control study. Bmj. 2011; 343:d5228. Epub 2011/09/14. <u>https://doi.org/10.1136/bmj.d5228</u> PMID: <u>21911446</u>; PubMed Central PMCID: PMC3171211.
- Antoniou T, Hollands S, Macdonald EM, Gomes T, Mamdani MM, Juurlink DN. Trimethoprim-sulfamethoxazole and risk of sudden death among patients taking spironolactone. Cmaj. 2015; 187(4): E138–43. Epub 2015/02/04. https://doi.org/10.1503/cmaj.140816 PMID: 25646289; PubMed Central PMCID: PMC4347789.

- Fralick M, Macdonald EM, Gomes T, Antoniou T, Hollands S, Mamdani MM, et al. Co-trimoxazole and sudden death in patients receiving inhibitors of renin-angiotensin system: population based study. Bmj. 2014; 349:g6196. Epub 2014/11/02. https://doi.org/10.1136/bmj.g6196 PMID: 25359996; PubMed Central PMCID: PMC4214638.
- Crellin E, Mansfield KE, Leyrat C, Nitsch D, Douglas IJ, Root A, et al. Trimethoprim use for urinary tract infection and risk of adverse outcomes in older patients: cohort study. 2018. <u>https://doi.org/10.1136/ bmj.k341</u> PMID: 29438980
- 20. Herrett E, Gallagher AM, Bhaskaran K, Forbes H, Mathur R, van Staa T, et al. Data Resource Profile: Clinical Practice Research Datalink (CPRD). Int J Epidemiol. 2015; 44(3):827–36. Epub 2015/06/08. https://doi.org/10.1093/ije/dyv098 PMID: 26050254; PubMed Central PMCID: PMC4521131.
- Benchimol EI, Smeeth L, Guttmann A, Harron K, Moher D, Petersen I, et al. The REporting of studies Conducted using Observational Routinely-collected health Data (RECORD) statement. PLoS Med. 2015; 12(10):e1001885. Epub 2015/10/07. https://doi.org/10.1371/journal.pmed.1001885 PMID: 26440803; PubMed Central PMCID: PMC4595218.
- Levey AS, Coresh J, Greene T, Stevens LA, Zhang YL, Hendriksen S, et al. Using standardized serum creatinine values in the modification of diet in renal disease study equation for estimating glomerular filtration rate. Ann Intern Med. 2006; 145(4):247–54. Epub 2006/08/16. PMID: 16908915.
- Khan NF, Perera R, Harper S, Rose PW. Adaptation and validation of the Charlson Index for Read/ OXMIS coded databases. BMC Family Practice. 2010; 11(1):1. doi: info:pmid/20051110.
- Austin PC, Grootendorst P, Anderson GM. A comparison of the ability of different propensity score models to balance measured variables between treated and untreated subjects: a Monte Carlo study. Stat Med. 2007; 26(4):734–53. Epub 2006/05/19. https://doi.org/10.1002/sim.2580 PMID: 16708349.
- Austin PC. Balance diagnostics for comparing the distribution of baseline covariates between treatment groups in propensity-score matched samples. Stat Med. 2009; 28(25):3083–107. Epub 2009/09/17. https://doi.org/10.1002/sim.3697 PMID: 19757444; PubMed Central PMCID: PMC3472075.
- Hsiao CY, Yang HY, Hsiao MC, Hung PH, Wang MC. Risk Factors for Development of Acute Kidney Injury in Patients with Urinary Tract Infection. PLoS ONE. 2015; 10(7). <u>https://doi.org/10.1371/journal.pone.0133835</u> PMID: 26213991.
- 27. Singh N, Gandhi S, McArthur E, Moist L, Jain AK, Liu AR, et al. Kidney function and the use of nitrofurantoin to treat urinary tract infections in older women. Cmaj. 2015; 187(9):648–56. Epub 2015/04/29. https://doi.org/10.1503/cmaj.150067 PMID: 25918178; PubMed Central PMCID: PMC4467927.
- **28.** England PH. English Surveillance Programme for Antimicrobial Utilisation and Resistance (ESPAUR). 2017.
- Muller AE, Verhaegh EM, Harbarth S, Mouton JW, Huttner A. Nitrofurantoin's efficacy and safety as prophylaxis for urinary tract infections: a systematic review of the literature and meta-analysis of controlled trials. Clin Microbiol Infect. 2017; 23(6):355–62. Epub 2016/08/21. https://doi.org/10.1016/j.cmi. 2016.08.003 PMID: 27542332.
- Huttner A, Verhaegh EM, Harbarth S, Muller AE, Theuretzbacher U, Mouton JW. Nitrofurantoin revisited: a systematic review and meta-analysis of controlled trials. J Antimicrob Chemother. 2015; 70 (9):2456–64. Epub 2015/06/13. https://doi.org/10.1093/jac/dkv147 PMID: 26066581.
- Whiting P, Morden A, Tomlinson LA, Caskey F, Blakeman T, Tomson C, et al. What are the risks and benefits of temporarily discontinuing medications to prevent acute kidney injury? A systematic review and meta-analysis. 2017. https://doi.org/10.1136/bmjopen-2016-012674 PMID: 28389482